Author's response to reviews

Title: Screening of exon 11 for BRCA1 using the High Resolution melting for diagnostic in Moroccan breast cancer patients

Authors:

Meryam El Khachibi (el.khachibi.meryam@gmail.com)
Khalil Hamzi (khalil.hamzi@gmail.com)
Brehima Diakite (br.diakite@yahoo.fr)
Mohamed A Senhaji (m.amine.senhaji@gmail.com)
Amina BAKHCHANE (amina_bakhchane@yahoo.fr)
Amal Tazzite (amal_tazzite@yahoo.fr)
Hassan Jouhadi (hjouhadi@yahoo.fr)
Abdelhamid Barakat (hamid.barakat@pasteur.ma)
Abdellatif Benieder (benider.abdel@yahoo.fr)
Sellama Nadifi (nadifisel@yahoo.fr)

Version: 2
Date: 9 May 2014

Author's response to reviews: see over
Covering Letter

To the Editor

Submission of Manuscript for publication

Dear Sir,

We intend to publish an article entitled “Screening of exon 11 for BRCA1 using the High Resolution melting for diagnostic in Moroccan breast cancer patients” in your esteemed journal as an Original article. On behalf of all the contributors, I will act guarantor and will correspond with the journal from this point onward prior to publication supported by the Hassan II academy. No conflicts of interest, permission for reproducing. We hereby transfer, assign, or otherwise convey all copyright ownership, including any and all rights incidental thereto, exclusively to the journal, in the event that such work is published by the journal.

Thanking you,

Yours’ Sincerely,

Corresponding Author: Sellama Nadifi, Genetics and Molecular Pathology Laboratory

Medical School of Casablanca, 19 Rue Tarik Ibnou Ziad, BP. 9154, Tel: 00212661092201

Email: nadifi@labgenmed.com